Author's response to reviews

**Title:** Physical activity, cardiorespiratory fitness, and metabolic syndrome in adolescents: a cross-sectional study.

**Authors:**

Antonio Stabelini Neto (asneto@uenp.edu.br)
Rodrigo Bozza (rdbozza@gmail.com)
Anderson Z Ulbrich (anderson_u@hotmail.com)
Luis Paulo G Mascarenhas (luismsk@gmail.com)
Margaret Cristina S Boguszewski (margabogus@uol.com.br)
Sergio G da Silva (sergiogregorio@ufpr.br)
Jeffer E Sasaki (jeffersasaki@gmail.com)
Wagner D Campos (wagner@ufpr.br)

**Version:** 5  **Date:** 23 August 2011

**Author's response to reviews:**

All formatting changes were made in accordance with the editor request.